Author's response to reviews

Title: Smoking, alcohol and dietary choices: evidence from the Portuguese National Health Survey

Authors:

Patricia Padrao (patriciapadrao@fcna.up.pt)
Nuno Lunet (nlunet@med.up.pt)
Ana-Cristina Santos (acsantos@med.up.pt)
Henrique Barros (hbarros@med.up.pt)

Version: 6 Date: 29 June 2007

Author's response to reviews: see over
Porto, June 29th 2007

Dear Dr. Lolu da Silva,

We appreciate the opportunity to have our manuscript published at BMC Public Health. As suggested, we have asked a professional copyediting service (www.sciencemanager.com), to check the manuscript language. The formatting changes were also made as required. We have corrected some errors found in tables 3 and 4.

We hope that the manuscript is now ready for acceptance.

Yours sincerely,

Patrícia Padrão and Nuno Lunet